Author’s response to reviews

Title: CD133 expression is not an independent prognostic factor in stage II and III colorectal cancer but may predict the better outcome in patients with adjuvant therapy

Authors:

Khalilullah Mia Jan Dr (khalilullahkliwal@googlemail.com)
So Young Jung MS (linas21@nate.com)
Ik-Yong Kim MD, PhD (iykim@yonsei.ac.kr)
Sung Soo Oh MD, PhD (oss0609@yonsei.ac.kr)
EunHee Choi PhD (eunhee_choi@yonsei.ac.kr)
Sei Jin Chang PhD (chang0343@yonsei.ac.kr)
Tae Young Kang MT (itty-95@hanmail.net)
Mee-Yon Cho MD, PhD (meeyon@yonsei.ac.kr)

Version: 5 Date: 17 September 2012

Author’s response to reviews:

Dear Editor-in Chief,

It has been a pleasure to hear from you and feel assured that our research work has been considered for peer reviewing. As per your recommendation, we have included in brief statement about the ethical conduct of the very significant steps of the research work which has been carried out in accordance with declaration of Helsinki and regulation of IRB committee. You can find this sentence with approval number in the Material and Methods section of the revised manuscript.

We are looking forward to hearing from you about the progress of peer review and other relevant subjects.

Sincerely,
Mee-Yon Cho, MD, PhD